Author's response to reviews

Title: Quitline Referral vs. Self-help Manual for Tobacco Use Cessation in the Emergency Department: A Feasibility Study

Authors:

Nicola E E Schiebel (schiebel.nicola@mayo.edu)
Jon O Ebbert (ebbert.jon@mayo.edu)

Version: 4 Date: 31 May 2007

Author's response to reviews: see over
Dear Editors

We would like to thank the reviewers for the time they spent reviewing our manuscript and for their helpful comments which, we believe, significantly improve the manuscript. We would also like to thank you for the opportunity to provide these edits and resubmit. We have addressed all the concerns of the third reviewer. We have changed the title of the manuscript as suggested and clarified the discussion and conclusions. Throughout the manuscript we have noted changes addressing reviewer suggestions in red, and highlighted objectives in bold to clarify limitations of the study. Significant deletions are noted with “track changes” deletion notation from word program functions.

Please do not hesitate to contact me with any questions.

Sincerely,

Nicola E. E. Schiebel, MD
Assistant Professor
Mayo Clinic College of Medicine